Author's response to reviews

Title: Diagnostic imaging equivalence testing comparing the Ocelot and the Dragonfly Optical Coherence Tomography Systems in an In-Vitro Superficial Femoral Artery Model

Authors:

Suhail Dohad (Suhail.dohad@cshs.org)
John Shao (John.Shao@gmail.com)
Ian Cawich (Ian.Cawich@gmail.com)
Manish Kankaria (mkankaria@avinger.com)
Arjun Desai (jj@avinger.com)

Version: 2
Date: 4 March 2015

Author's response to reviews:

Updating according to Editors request